Author’s response to reviews

Title: Flow-cytometric monitoring of disease-associated expression of 9-O-acetylated sialoglycoproteins in combination with known CD antigens, as an index for MRD in children with acute lymphoblastic leukaemia: a two-year longitudinal follow-up study

Authors:

Suchandra Chowdhury (cmandal@iicb.res.in)
Suman Bandyopadhyay (cmandal@iicb.res.in)
Chandan Mandal (cmandal@iicb.res.in)
Sarmila Chandra (cmandal@iicb.res.in)
Chitra Mandal (cmandal@iicb.res.in)

Version: 4 Date: 28 January 2008

Author’s response to reviews:

To,

The Editor, BMC Cancer, Attention to Dr. Mona Jazayeri, Assistant editor,
BMC-series journals,
BioMed Central Ltd, Middlesex House, 34-42 Cleveland Street, London W1T 4LB, UK
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Sub: Submission of manuscript 3806764931566294 after incorporating the formatting changes

Dear Sir or Madam,

Let me take this opportunity to thank you for accepting our manuscript. As per your suggestion the changes have been incorporated in the manuscript 3806764931566294.

With best regards,

Yours sincerely,

Dr. Chitra Mandal